Author's response to reviews

Title: Determination of Interleukin-6 and Tumor Necrosis Factor-alpha concentrations in Iranian-Khorasanian patients with preeclampsia.

Authors:

Jalil Tavakkol Afshari (jtavakkol@yahoo.com)
Ahmad Shameli (Ahmad-shameli@mums.ac.ir)
Mahmood Shakeri (M-shakeri@mums.ac.ir)
Mohammad A Fahmidehkar (mafahmidehkar@yahoo.com)
Nahid Ghomian (N-ghomian@mum.ac.ir)
Maryam Mahajer (mmahajer@yahoo.com)
Mehdirezka Emadzadeh (M-emadzadeh@mums.ac.ir)
Rasoul Khoshnavaz (Rasoulkhoshnavazmd@yahoo.com)

Version: 4 Date: 26 August 2005

Author's response to reviews:

Dear Dr. Holmes,

Thank you for approving of our manuscript for publication in your journal. We have reviewed our manuscript according to the BMC pregnancy and childbirth.

The following has been changed:

Affiliation:

J. Tavakkol Afshari1, N. Ghomian2, A. Shameli1, M. T. Shakeri3, M. A. Fahmidehkar1, M. Mahajer1, R. Khoshnavaz1, and M. Emadzadeh4

1 Immunogenetics department, Immunology Research Group, Bu-Ali Research Institute, Bu-Ali Sq., Mashhad University of Medical Sciences (MUMS), Mashhad, Iran
2 Obstetrics & Gynecology department, Imam Reza Hospital, Imam Reza Sq., Mashhad University of Medical Sciences (MUMS), Mashhad, Iran
3 Community medicine department, Medical school, Daneshgah St., Mashhad University of Medical Science (MUMS), Mashhad-Iran
Department of Cardiology, Ghaem Hospital, Ahmad Abad St., Mashhad University of Medical Sciences (MUMS), Mashhad, Iran

Highlighting/Tracking:

Highlighting (As red colour were removed).

Introduction:

renamed this section 'Background'.

References:

have been cited in the text using consecutive numbers in square brackets.

Materials & Methods:
has been rename this section 'Methods'.

References:
The reference list has been formatted in the journal's style.

Figure title:
The legends have been added to the manuscript.

yours Sincerely,

Tavakkol Afshari, MSc, PhD, MT(ASCP), CLS(NCA)
Associate professor of Medicine